Author's response to reviews

Title: Re-organisation of oesophago-gastric cancer care in England: progress and remaining challenges

Authors:

Tom R Palser (tpalser@rcseng.ac.uk)
David A Cromwell (david.cromwell@lshtm.ac.uk)
Richard H Hardwick (richard.hardwick@addenbrookes.nhs.uk)
Stuart A Riley (Stuart.Riley@sth.nhs.uk)
Kimberley Greenaway (Kimberley.Greenaway@ic.nhs.uk)
William Allum (wmallum@hotmail.com)
Jan HP van der Meulen (jan.vandermeulen@lshtm.ac.uk)

Version: 6 Date: 26 October 2009

Author's response to reviews:

Changes have been made to the document as described in the email from the BMC Editorial Production Team. In particular

1. Affiliations have been stated in full.
2. All authors have been added to the citations in the reference list
3. Table 1 and Box 1 have been reformatted. Box 1 is now labelled Table 6.
4. Information on the additional files have been added and are referred to in the text
5. The acknowledgements have been reorganised.